Author’s response to reviews

Title: Dietary zinc intake and its determinants among Ethiopian children 6-35 months of age

Authors:

Girmay Mersha (girmayayana@yahoo.com)
Tibebu Moges (tibebumoges@gmail.com)
Aregash Samuel (aregashhafebo@gmail.com)
Tsehai Assefa (tsehaiassefa@yahoo.com)
Solomon Eshetu (Solomoneshetu13@yahoo.com)
Aweke Kebede (awekekeb@gmail.com)

Version: 2 Date: 21 May 2018

Author’s response to reviews:

Date: 22 May 2018

TO: The Editor

BMC Nutrition

Dear Sir or Madam:

We would like to take this opportunity to thank you for potentially accepting our manuscript for publication in your journal and we appreciate your comments and suggestions which certainly improved our manuscript entitled “Dietary zinc intake and its determinants among Ethiopian children 6-35 months of age”.

As requested, we have addressed all comments made by the editor and reviewers, revised the manuscript and provided you with point by point response for the comments made. We hope this satisfies your request and we look forward to you accepting our manuscript and for its publication in your journal.

With best regards

Girmay Ayana

Corresponding Author